<%namespace name="tools" file="tools.tex"/>

\section{Small scale laboratory tests}

\subsection{M1263 I: Wind Flume 1974-1975}

The results depicted in \autoref{fig:Windflume_M1263_I_AT33_final_profiles} to \autoref{fig:Wflume_M1263_I_DT98_final_profiles} were part of a series of experiments performed during 1974 and 1975 in the Wind Flume of Laboratory De Voorst in The Netherlands\citep{VanDeGraaff1976}. During the experiments, depth scales of 84, 47 and 26 were used. 

${tools.image('Windflume_M1263_I/AT33/', 'final_profiles.png', 'Final profile of test AT33')}
${tools.image('Windflume_M1263_I/AT47/', 'final_profiles.png', 'Final profile of test AT47')}
${tools.image('Windflume_M1263_I/AT61/', 'final_profiles.png', 'Final profile of test AT61')}
${tools.image('Windflume_M1263_I/AT71/', 'final_profiles.png', 'Final profile of test AT71')}
${tools.image('Windflume_M1263_I/AT91/', 'final_profiles.png', 'Final profile of test AT91')}
${tools.image('Windflume_M1263_I/AT95/', 'final_profiles.png', 'Final profile of test AT95')}
${tools.image('Windflume_M1263_I/BT13/', 'final_profiles.png', 'Final profile of test BT13')}
${tools.image('Windflume_M1263_I/BT15/', 'final_profiles.png', 'Final profile of test BT15')}
${tools.image('Windflume_M1263_I/BT17/', 'final_profiles.png', 'Final profile of test BT17')}
${tools.image('Windflume_M1263_I/BT23/', 'final_profiles.png', 'Final profile of test BT23')}
${tools.image('Windflume_M1263_I/BT25/', 'final_profiles.png', 'Final profile of test BT25')}
${tools.image('Windflume_M1263_I/BT27/', 'final_profiles.png', 'Final profile of test BT27')}
${tools.image('Windflume_M1263_I/BT45/', 'final_profiles.png', 'Final profile of test BT45')}
${tools.image('Windflume_M1263_I/BT62/', 'final_profiles.png', 'Final profile of test BT62')}
${tools.image('Windflume_M1263_I/BT72/', 'final_profiles.png', 'Final profile of test BT72')}
${tools.image('Windflume_M1263_I/BT92/', 'final_profiles.png', 'Final profile of test BT92')}
${tools.image('Windflume_M1263_I/BT96/', 'final_profiles.png', 'Final profile of test BT96')}
${tools.image('Windflume_M1263_I/CT14/', 'final_profiles.png', 'Final profile of test CT14')}
${tools.image('Windflume_M1263_I/CT16/', 'final_profiles.png', 'Final profile of test CT16')}
${tools.image('Windflume_M1263_I/CT18/', 'final_profiles.png', 'Final profile of test CT18')}
${tools.image('Windflume_M1263_I/CT24/', 'final_profiles.png', 'Final profile of test CT24')}
${tools.image('Windflume_M1263_I/CT26/', 'final_profiles.png', 'Final profile of test CT26')}
${tools.image('Windflume_M1263_I/CT28/', 'final_profiles.png', 'Final profile of test CT28')}
${tools.image('Windflume_M1263_I/CT46/', 'final_profiles.png', 'Final profile of test CT46')}
${tools.image('Windflume_M1263_I/CT63/', 'final_profiles.png', 'Final profile of test CT63')}
${tools.image('Windflume_M1263_I/CT73/', 'final_profiles.png', 'Final profile of test CT73')}
${tools.image('Windflume_M1263_I/CT93/', 'final_profiles.png', 'Final profile of test CT93')}
${tools.image('Windflume_M1263_I/CT97/', 'final_profiles.png', 'Final profile of test CT97')}
${tools.image('Windflume_M1263_I/DT34/', 'final_profiles.png', 'Final profile of test DT34')}
${tools.image('Windflume_M1263_I/DT48/', 'final_profiles.png', 'Final profile of test DT48')}
${tools.image('Windflume_M1263_I/DT64/', 'final_profiles.png', 'Final profile of test DT64')}
${tools.image('Windflume_M1263_I/DT74/', 'final_profiles.png', 'Final profile of test DT74')}
${tools.image('Windflume_M1263_I/DT94/', 'final_profiles.png', 'Final profile of test DT94')}
${tools.image('Windflume_M1263_I/DT98/', 'final_profiles.png', 'Final profile of test DT98')}

\subsection{M1263 II: Wind Flume 1976-1977}

\autoref{fig:Windflume_M1263_II_101_final_profiles} to \autoref{fig:Windflume_M1263_II_128_final_profiles} depict the tests performed in part II of the M1263 experiments, also in the Wind Flume and with depth scales of 84, 47 and 26\citep{Vellinga1981}. 

${tools.image('Windflume_M1263_II/101/', 'final_profiles.png', 'Final profile of test 101')}
${tools.image('Windflume_M1263_II/105/', 'final_profiles.png', 'Final profile of test 105')}
${tools.image('Windflume_M1263_II/111/', 'final_profiles.png', 'Final profile of test 111')}
${tools.image('Windflume_M1263_II/115/', 'final_profiles.png', 'Final profile of test 115')}
${tools.image('Windflume_M1263_II/121/', 'final_profiles.png', 'Final profile of test 121')}
${tools.image('Windflume_M1263_II/122/', 'final_profiles.png', 'Final profile of test 122')}
${tools.image('Windflume_M1263_II/123/', 'final_profiles.png', 'Final profile of test 123')}
${tools.image('Windflume_M1263_II/124/', 'final_profiles.png', 'Final profile of test 124')}
${tools.image('Windflume_M1263_II/125/', 'final_profiles.png', 'Final profile of test 125')}
${tools.image('Windflume_M1263_II/126/', 'final_profiles.png', 'Final profile of test 126')}
${tools.image('Windflume_M1263_II/127/', 'final_profiles.png', 'Final profile of test 127')}
${tools.image('Windflume_M1263_II/128/', 'final_profiles.png', 'Final profile of test 128')}

\subsection{M1819 I: Scheldt Flume 1981}

In 1981, small scale dune erosion experiments (with a depth scale of 30) were performed in the Scheldt Flume of Laboratory De Voorst in The Netherlands\citep{Tilmans1982}, the results of which are shown in \autoref{fig:Scheldtflume_M1819_I_T01_final_profiles} to \autoref{fig:Scheldtflume_M1819_I_T29_final_profiles}.

${tools.image('Scheldtflume_M1819_I/T01/', 'final_profiles.png', 'Final profile of test T01')}
${tools.image('Scheldtflume_M1819_I/T02/', 'final_profiles.png', 'Final profile of test T02')}
${tools.image('Scheldtflume_M1819_I/T03/', 'final_profiles.png', 'Final profile of test T03')}
${tools.image('Scheldtflume_M1819_I/T04/', 'final_profiles.png', 'Final profile of test T04')}
${tools.image('Scheldtflume_M1819_I/T05/', 'final_profiles.png', 'Final profile of test T05')}
${tools.image('Scheldtflume_M1819_I/T06/', 'final_profiles.png', 'Final profile of test T06')}
${tools.image('Scheldtflume_M1819_I/T07/', 'final_profiles.png', 'Final profile of test T07')}
${tools.image('Scheldtflume_M1819_I/T08/', 'final_profiles.png', 'Final profile of test T08')}
${tools.image('Scheldtflume_M1819_I/T09/', 'final_profiles.png', 'Final profile of test T09')}
${tools.image('Scheldtflume_M1819_I/T10/', 'final_profiles.png', 'Final profile of test T10')}
${tools.image('Scheldtflume_M1819_I/T11/', 'final_profiles.png', 'Final profile of test T11')}
${tools.image('Scheldtflume_M1819_I/T12/', 'final_profiles.png', 'Final profile of test T12')}
${tools.image('Scheldtflume_M1819_I/T13/', 'final_profiles.png', 'Final profile of test T13')}
${tools.image('Scheldtflume_M1819_I/T14/', 'final_profiles.png', 'Final profile of test T14')}
${tools.image('Scheldtflume_M1819_I/T21/', 'final_profiles.png', 'Final profile of test T21')}
${tools.image('Scheldtflume_M1819_I/T22/', 'final_profiles.png', 'Final profile of test T22')}
${tools.image('Scheldtflume_M1819_I/T23/', 'final_profiles.png', 'Final profile of test T23')}
${tools.image('Scheldtflume_M1819_I/T24/', 'final_profiles.png', 'Final profile of test T24')}
${tools.image('Scheldtflume_M1819_I/T25/', 'final_profiles.png', 'Final profile of test T25')}
${tools.image('Scheldtflume_M1819_I/T26/', 'final_profiles.png', 'Final profile of test T26')}
${tools.image('Scheldtflume_M1819_I/T27/', 'final_profiles.png', 'Final profile of test T27')}
${tools.image('Scheldtflume_M1819_I/T28/', 'final_profiles.png', 'Final profile of test T28')}
${tools.image('Scheldtflume_M1819_I/T29/', 'final_profiles.png', 'Final profile of test T29')}

\subsection{H4265: Scheldt Flume 2003}

During 2003, Scheldt Flume experiments were performed to assess the influence of the wave period on the amount of dune erosion. The tests are performed using a depth scale of 30, resulting in the profiles depicted in \autoref{fig:Scheldtflume_H4265_T01_final_profiles} to \autoref{fig:Scheldtflume_H4265_T13_final_profiles}.

${tools.image('Scheldtflume_H4265/T01/', 'final_profiles.png', 'Final profile of test T01')}
${tools.image('Scheldtflume_H4265/T02/', 'final_profiles.png', 'Final profile of test T02')}
${tools.image('Scheldtflume_H4265/T02a/', 'final_profiles.png', 'Final profile of test T02a')}
${tools.image('Scheldtflume_H4265/T03/', 'final_profiles.png', 'Final profile of test T03')}
${tools.image('Scheldtflume_H4265/T11/', 'final_profiles.png', 'Final profile of test T011')}
${tools.image('Scheldtflume_H4265/T12/', 'final_profiles.png', 'Final profile of test T012')}
${tools.image('Scheldtflume_H4265/T13/', 'final_profiles.png', 'Final profile of test T013')}


